Author's response to reviews

Title: Percutaneous septal ablation for left mid-ventricular obstructive hypertrophic cardiomyopathy: a case report

Authors:

Istemihan Tengiz (dritengiz@yahoo.com)
Ertugrul Ercan (ertugrulercan@yahoo.com)
Emin Alioglu (emilaliyev@mynet.com)
Ugur O. Turk (droturk@yahoo.com)

Version: 6 Date: 14 December 2005

Author's response to reviews:

Dear Editor,
We have just revised our manuscript and requiring changes have been made.
Keywords section removed from the manuscript.
Competing interests' and Authors' contributions sections included in the manuscript.
Best regards,
Dr I. Tengiz